Editorial board: World Journal of Hepatology

Dear Professors Ke-Qin Hu, Koo Jeong Kang and Nikolaos Pyrsopoulos – Editors-in-Chief

I am herewith submitting the revised manuscript entitled: “Current aspects of renal dysfunction after liver transplantation” for appreciation of the World Journal of Hepatology Editorial Board, from authors Mariana P Pacheco, Luiz Augusto Carneiro-D’Albuquerque and Daniel F. Mazo.

We are grateful to the reviewer for the helpful comments on the original version of our manuscript. We have taken all the comments into account in order to resubmit this revised version with improvements. Please find our answers to all the queries below.

The manuscript was previously sent to copyediting service by a professional English language editing company (American Manuscript Editors), (certificate attached).

This is an extensive review of renal dysfunction after liver transplantation, with several references to relevant and specific studies (originally 138 references), in accordance with the journal's recommendations at the time of manuscript preparation. Many references were removed from this revised version, following the Science Editor suggestion. However, the limitation of up to three citations from the same journal could have compromised the quality of the manuscript and has not been possible for every occurrence.

We hope the revised version of our paper might now be suitable for publication in World Journal of Hepatology and we look forward to hearing from you at your earliest convenience.

Thank you in advance,

Daniel Ferraz de Campos Mazo, MD, PhD
Department of Gastroenterology- University of Sao Paulo School of Medicine
Av. Dr. Enéas de Carvalho Aguiar nº 255, Instituto Central, # 9159,
05403-900 - Sao Paulo – Brazil. Phone: 55-11-2661-6447
Fax: 55-11-2661-7830. E-mail: daniel.mazo@hc.fm.usp.br

Av. Dr. Enéas de Carvalho Aguiar nº 255, Instituto Central, # 9159 - 05403-900
Sao Paulo - Brazil
Reviewers' comments:
Reviewer #1:

I have read the manuscript “Current aspects of renal dysfunction after liver transplantation” with interest. This review is extensive but there are some comments.

Questions / comments:

1- There should be a graphic summary of risk and management, this will help more citation of this review.
Response: A graphic summary was provided, along with Figures archive.

2- Abbreviation likes LT has been used, “Transplantation” should be “LT” instead.
Response: This recommendation was followed, according to the reviewer's suggestion.